Author's response to reviews

Title: Temporal trend and climate factors of hemorrhagic fever with renal syndrome epidemic in Shenyang City, China

Authors:

Xiaodong Liu (liuxd1983@126.com)
Baofa Jiang (bjiang@sdu.edu.cn)
Weidong Gu (wgu@uab.edu)
Qiyong Liu (liuqiyong@icdc.cn)

Version: 5 Date: 8 November 2011

Author's response to reviews: see over
Dear Editors,

Thanks very much for your kind work and consideration on publication of our paper entitled “Temporal Trend and Climate Factors of Hemorrhagic Fever with Renal Syndrome Epidemic in Shenyang City, China” (ID: 2903271448538366). On behalf of my co-authors, we would like to express our great appreciation to editor and reviewers. Thank you and best regards.

We have checked the current version of our manuscript carefully and have made some minor correction which we hope meet with approval.

1. Figure 4:

   We think that Figure 4 is need not to change. Because the X-axis labels are stay in their corresponding time interval of actual value. Some variables are correlated with the monthly HFRS cases during the previous 3 months, so the fitted values are begin from March 2004.

2. ”Cross Cross-correlation analyses” was corrected as “Cross-correlation analysis”

Yours sincerely,

Xiaodong Liu